Author's response to reviews

Title: Association of decreased mitochondrial DNA content with colorectal cancer progression

Authors:

haihong cui (hh_cui@sina.com)
ping huang (hp_20050909@126.com)
ying han (hanying@sina.com)
yunxin zhang (zhangyx@sina.com)
zhenhua zhang (zhangzh@sina.com)
wei xu (18910998563@189.cn)
xiaopeng wang (wangxp@126.com)

Version: 5 Date: 19 February 2013

Author's response to reviews: see over
Reviewer's report
Title: Association of decreased mitochondrial DNA content with colorectal cancer progression
Version: 4 Date: 10 January 2013
Reviewer: Hsin-Chen Lee
Reviewer's report:
Minor Essential Revisions
Table 1 contains a wrong number in the group of poorly differentiated.

The original number of the table is wrong, when I was revising the manuscript, I found the mistake, and changed it.

Level of interest: An article whose findings are important to those with closely related research interests
Quality of written English: Acceptable
Statistical review: No, the manuscript does not need to be seen by a statistician.
Declaration of competing interests:
I declare that I have no competing interests.